Author's response to reviews

Title: Difference in immune response in vaccinated and unvaccinated Swedish individuals after the 2009 Influenza pandemic

Authors:

Isabelle Magalhaes (Isabelle.Magalhaes@ki.se)
Mikael Eriksson (Mikael.Eriksson@ki.se)
Charlotte Linde (charlottelinde@gmail.com)
Rashid Muhammad (mura0018@student.umu.se)
Lalit Rane (lalit.rane@ki.se)
Aditya Ambati (Aditya.Ambati@ki.se)
Rebecca Axelsson-Robertsson (Rebecca.Axelsson@ki.se)
Bahareh Khalaj (bahr_kh2003@yahoo.com)
Nancy Alvarez (montseac26@yahoo.com)
Giulia Lapini (lapini7@unisi.it)
Emanuele Montomoli (emanuele.montomoli@unisi.it)
Annika Linde (Annika.Linde@smi.se)
Nancy Pedersen (Nancy.Pedersen@ki.se)
Markus Maeurer (Markus.Maeurer@ki.se)

Version: 5
Date: 8 May 2014

Author's response to reviews:

Dear Editors,

we have added the information concerning the ethics committee (Ethics committee Stockholm south review board (DN 2009/1183-31) and hope that the report is acceptable for publication.

For the authors,
Mark Maeurer